Author’s response to reviews

Title: Simultaneous splenic mantle cell lymphoma and bone marrow marginal zone lymphoma with peripheral expression. A rare case of discordant lymphoma

Authors:

Giovanni Carulli (giovannicarulli@alice.it)
Alessandra Marini (alessandra.marini@hotmail.it)
Eugenio M Ciancia (e.ciancia@ao-pisa.toscana.it)
Josef Bruno (j.bruno@usl12.toscana.it)
Silvana Vignati (s.vignati@ao-pisa.toscana.it)
Paola Lambelet (p.lambelet@usl12.toscana.it)
Elisa Cannizzo (elisacannizzo@libero.it)
Virgina Ottaviano (virotta77@libero.it)
Sara Galimberti (s.galimberti@med.unipi.it)
Francesco Caracciolo (f.caracciolo@med.unipi.it)
Maria I Ferreri (m.ferreri@ao-pisa.toscana.it)
Elena Ciabatti (elenaciabatti@yahoo.it)
Mario Petrini (m.petrini@med.unipi.it)

Version: 3 Date: 5 July 2011

Author’s response to reviews: see over
To the Editor of the Journal of Medical Case Reports

Dear Sir,

with reference to our manuscript MS: 1331065085500659 entitled “Simultaneous splenic mantle cell lymphoma and bone marrow marginal zone lymphoma with peripheral expression. A rare case of discordant lymphoma”, please find enclose a new version after minor changes, as suggested.

The following new changes were made:

1) patient's ethnicity has been included in the Abstract;
2) sub-headings have been removed and the methods section has been merged in the case presentation section.

All changes made after the first revision still are underlined.

Thank you very much for your kind attention.

Sincerely, Giovanni Carulli MD